Author's response to reviews

Title: Adenoma of Vater Ampulla. A case report study

Authors:

Konstantinos Alexiou (knegro@otenet.gr)
Argyrios Ioannidis (argioannid@med.uoa.gr)
Ioannis Drikos (johndrikos@yahoo.com)
Athanasios Fotopoulos (athfot@gmail.com)
Ioannis Karanikas (ikaranikas@yahoo.gr)
Nikolaos Economou (gramdiy@sismanoglio.gr)

Version: 5
Date: 27 February 2014

Author's response to reviews:

1. Please include a Competing Interests section before the Reference list. If the authors have no competing interests, please state: "The authors declare that they have no competing interests."

Done

2. Please remove the date figure 2 was taken as this may jeopardize the anonymity of the patient.

Done